Author’s response to reviews

Title: Paracrine effects of CCN3 from non-cancerous hepatic cells increase signaling and progression of hepatocellular carcinoma

Authors:

Qingan Jia (qajia66@163.com)
Weimin Li (760230249@qq.com)
Xia Liao (373151023@qq.com)
Pengbo Ning (127858600@qq.com)
Yu Cao (18792629638@163.com)
Mei Zhang (xazhangmei@126.com)
Yang Bu (boyang1976@163.com)
Jun Lv (junlv0088@163.com)

Version: 2 Date: 10 Dec 2018

Author’s response to reviews:

The overlap in the manuscript was revised, and the language and document were also proof read by Biosciencewriters company in British English.